Author's response to reviews

Title: Persons with dementia missing in the community: Is it wandering or something unique?

Authors:

Meredeth A Rowe (mrowe@ufl.edu)
Sydney S Vandeveer (sydneyvand@gmail.com)
Greenblum A Catherine (cgreenblum@aol.com)
Cassandra N List (cnlist23@gmail.com)
Rachael M Fernandez (rachael.fernandez@me.com)
Natalie E Mixson (nemixson@gmail.com)
Hyo C Ahn (hcahn@ufl.edu)

Version: 3 Date: 23 May 2011

Author's response to reviews:

Formatting changes to be made:
-------------------------------

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system
*the title in the submission system was updated

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
* corrected to match in both places

Abstract: This should be identical in the manuscript and the submission system.
*corrected in submission system

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
*All authors have read and approved the manuscript.

References: Please make sure references are cited in numerical order. We notice reference 21 is cited before 20 in the text.
* corrected

References: reference 21, is missing from the reference list please provide details whilst ensuring references are still cited in consecutive order.
* corrected

Tables: the first table referenced in the manuscript text should be labelled Table 1. The second, Table 2, etc.
*corrected

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables- please remove the visible vertical lines from your tables
*done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
*done